Author's response to reviews

Title: Age differences in mental health literacy

Authors:

Louise Farrer (louise.farrer@anu.edu.au)
Liana Leach (liana.leach@anu.edu.au)
Kathleen M Griffiths (kathy.griffiths@anu.edu.au)
Helen Christensen (helen.christensen@anu.edu.au)
Anthony F Jorm (ajorm@unimelb.edu.au)

Version: 6 Date: 18 April 2008

Author's response to reviews:

The formatting changes have been made.